Author's response to reviews

Title: Fagerstrom Test for Nicotine Dependence vs Heavy Smoking Index in a general population survey.

Authors:

Monica Perez-Rios (monica.perez.rios@usc.es)
Maria Isolina Santiago-Pérez (monica.perez.rios@usc.es)
Begona Alonso (monica.perez.rios@usc.es)
Alberto Malvar (monica.perez.rios@usc.es)
Xurxo Hervada (monica.perez.rios@usc.es)
Jose de Leon (monica.perez.rios@usc.es)

Version: 5 Date: 26 November 2009

Author's response to reviews: see over
Dear Editor:

We have added tobacco labours definition, the ethics committee approval and informed consent in the methods section and also we have modified the contribution section. Thank you very much.

Sincerely yours

Mónica Pérez-Ríos